Author's response to reviews

Title: Influenza A/H1N1 septic shock in a patient with systemic lupus erythematosus. A case report

Authors:

Tselios Konstantinos (tselioskostas@yahoo.gr)
Tsioka Ritsa (agtsioka@yahoo.gr)
Sarantopoulos Alexandros (alexsar@med.auth.gr)
Mouloudi Eleni (elmoulou@yahoo.gr)
Boura Panagiota (boura@med.auth.gr)

Version: 5 Date: 15 December 2011

Author's response to reviews:

Dear Madam/Sir

I would like to inform you that
1. All minor changes were made according to the standard journal requirements.
2. A legend for Figure 1 is provided at the end of the manuscript.
3. The last phrase in the Author's Contribution section is appropriate.

I believe that all necessary revisions have now been accomplished. Thank you very much in advance.

Sincerely yours,

Konstantinos Tselios, MD